Author's response to reviews

Title: Linear accelerator-based stereotactic radiosurgery in 140 brain metastases from malignant melanoma

Authors:

Henrik Hauswald (henrik.hauswald@med.uni-heidelberg.de)
Alina Stenke (alina.stenke@gmail.com)
Juergen Debus (juergen.debus@med.uni-heidelberg.de)
Stephanie E Combs (stephanie.combs@med.uni-heidelberg.de)

Version: 2 Date: 3 October 2014

Author's response to reviews:

I am sorry for the Format Errors.

Fixed issues:
1. Line Numbering added
2. Title page: email of co-authors added.
3. Acknowledgements do not exist.
4. Ethics statement added ("The study was approved by the Ethics Committee of the University of Heidelberg.").